MATTEO RIDOLFI, IDLab, Department of Information Technology, Ghent University - imec ABDIL KAYA, IDLab, Faculty of Applied Engineering, University of Antwerp - imec RAFAEL BERKVENS, IDLab, Faculty of Applied Engineering, University of Antwerp - imec MAARTEN WEYN, IDLab, Faculty of Applied Engineering, University of Antwerp - imec WOUT JOSEPH, WAVES, Ghent University - imec ELI DE POORTER, IDLab, Department of Information Technology, Ghent University - imec

Ultra-Wideband (UWB) is a Radio Frequency (RF) technology that is currently used for accurate indoor localization. However, the cost of deploying such a system is large, mainly due to the need for manually measuring the exact location of the installed infrastructure devices ("anchor nodes"). Self-calibration of UWB reduces deployment costs because it allows for automatic updating of the coordinates of fixed nodes when they are installed or moved. Additionally, installation costs can also be reduced by using collaborative localization approaches where mobile nodes act as anchors. This article surveys the most significant research that has been done on self-calibration and collaborative localization. First, we find that often, these terms are improperly used, leading to confusion for the readers. Furthermore, we find that in most of the cases, UWB specific characteristics are not exploited, so that crucial opportunities to improve performance are lost. Our classification and analysis provide the basis for further research on self-calibration and collaborative localization and analysis provide the basis for further research tracks that are open for investigation and can lead to better performance, e.g. machine learning and optimized physical settings.

Additional Key Words and Phrases: survey, indoor localization, Ultra-Wideband, self-calibration, collaborative localization.

## ACM Reference Format:

Matteo Ridolfi, Abdil Kaya, Rafael Berkvens, Maarten Weyn, Wout Joseph, and Eli De Poorter. 2021. Self-calibration and Collaborative Localization for UWB Positioning Systems: A Survey and Future Research Directions. *ACM Comput. Surv.* 0, 0, Article 0 (2021), 28 pages. https://doi.org/0

#### **1 INTRODUCTION**

Location awareness in Radio Frequency (RF) systems provides a (mobile) node with its spatial coordinates, be it relative or absolute. This allows a variety of features such as location-based routing and scheduling, which are well researched

Authors' addresses: Matteo Ridolfi, matteo.ridolfi@ugent.be, IDLab, Department of Information Technology, Ghent University - imec, Technologiepark-Zwijnaarde 126, 9052 Gent, Belgium, Ghent, Belgium, 9000; Abdil Kaya, abdil kaya@uantwerpen.be, IDLab, Faculty of Applied Engineering, University of Antwerp - imec, Antwerp, Belgium, 2000; Rafael Berkvens, rafael.berkvens@uantwerpen.be, IDLab, Faculty of Applied Engineering, University of Antwerp - imec, Antwerp, Belgium, 2000; Maarten Weyn, maarten.weyn@uantwerpen.be, IDLab, Faculty of Applied Engineering, University of Antwerp - imec, Antwerp, Belgium, 2000; Maarten Weyn, maarten.weyn@uantwerpen.be, IDLab, Faculty of Applied Engineering, University of Antwerp - imec, Antwerp, Belgium, 2000; Wout Joseph, wout.joseph@ugent.be, WAVES, Ghent University - imec, Technologiepark-Zwijnaarde 126, 9052 Gent, Belgium, Ghent, Belgium, 9000; Eli De Poorter, eli.depoorter@ugent.be, IDLab, Department of Information Technology, Ghent University - imec, Technologiepark-Zwijnaarde 126, 9052 Gent, Belgium, Ghent, Belgium, 9000.

 $\circledast$  2021 Copyright held by the owner/author(s). Publication rights licensed to ACM. Manuscript submitted to ACM

Permission to make digital or hard copies of all or part of this work for personal or classroom use is granted without fee provided that copies are not made or distributed for profit or commercial advantage and that copies bear this notice and the full citation on the first page. Copyrights for components of this work owned by others than the author(s) must be honored. Abstracting with credit is permitted. To copy otherwise, or republish, to post on servers or to redistribute to lists, requires prior specific permission and/or a fee. Request permissions from permissions@acm.org.

topics for Wireless Sensor Networks (WSNs) [94, 119]. The diverse application scopes make these systems highly domain specific.

In the last couple of years, interest has grown in a particular technology called Ultra-Wideband (UWB) [30]. This is due to the availability of affordable UWB chips on the market. A rise that is culminating with the integration of UWB in one of the most popular consumer devices, i.e., the iPhone 11 from Apple. Additionally, UWB has been deeply researched and its accurate cm-level positioning capabilities are well known [106, 117]. Using this standard, position accuracy in the order of below 20 cm is possible. Unfortunately, nothing comes for free and UWB has a price to pay in terms of infrastructure installation costs and complexity, as well as robustness to non-line-of-sight (NLOS) conditions [26], which are typical features of indoor environments. This translates to problematic situations in which the fixed anchor nodes must be precisely surveyed for the system to properly work, and moreover, enough of them must be installed for good network coverage, i.e., all mobile nodes can communicate with the fixed ones.

In general, an UWB localization system is made of two types of node, i.e., the anchors and the tags. In the most frequently used anchor-based scenarios, the distances between the tags and the anchors must be known to pinpoint the position of the tags. Moreover, this calculation assumes the coordinates of the fixed anchor nodes to be already known to the system. This piece of information is normally retrieved by performing the so-called calibration procedure in which manual innervation is required to calculate the exact position of the fixed nodes. Several solutions [7, 18, 57, 83] show that for an average office size (in the order of 6x4 m<sup>2</sup>) with LOS conditions, four anchors should be sufficient to track hundreds of tags and therefore the installation process is usually smooth and cost effective. Once enough fixed nodes have been placed, the tags can estimate the distances to each of the anchors and calculate their own locations using algorithms such as multilateration.

Nonetheless, several scenarios involve completely different environments, and requirements. Other than office spaces, UWB indoor localization systems are typically used in huge warehouses, wherein covering the entire space would require a large amount of hardware (anchors) and manpower to position (calibrate) all the fixed nodes. Another case in which a standard installation like in an office space would be problematic is an emergency situation. Indeed, UWB technology can be used during a crisis [31], e.g., fire or rescuing interventions. Clearly in these scenarios, it is almost impossible to foresee an infrastructure and even if one can be deployed, its installation and calibration should be kept as easy as possible to minimize installation costs and time.

To cope with the various situations in which UWB systems can operate, we need targeted mechanisms designed to work in any condition. Self-calibration allows easy deployment of anchor nodes no matter in which space is done, regardless of any requirements and possible restrictions. The system automatically detects an anchor and computes its coordinates. This technique can be also used when anchors are moved to new locations to update their positions. Similarly, when anchors are already pinpointed to their exact location and cannot be moved or if installing new ones is not permitted, the system can make use of collaborative localization. The mobile tags can exchange messages not only with the anchors but also with other mobile nodes. This can be crucial in different situations as we will show in Section 5 about possible problems. Similarly to self-calibration, collaborative localization aims to reduce the costs of the overall UWB system by keeping the infrastructure costs as low as possible and allowing peer-to-peer communication, i.e., tag-to-tag. We find that these two techniques are extremely important to make the UWB systems more mature and robust. However, self-calibration and collaborative localization are often wrongly interchanged concepts in standard practice. Moreover, in the majority of papers we came across [8, 26, 33, 38, 55, 56, 75, 90, 109, 119], authors are often not clearly defining the problem, nor are they UWB specific, in the sense that even when the topics are surveyed, it is never clear how UWB fits into the problem and how its specific characteristics can be exploited. This aspect will carefully be Manuscript submitted to ACM

analyzed in the rest of the survey and in Section 6.3. We propose a classification that considers this and other aspects such as how information is distributed in the network and the type of algorithm adopted to position the nodes.

The main contributions of this paper are as follows:

- Solutions for both self-calibration and collaborative localization problems are rigorously surveyed. By emphasizing their differences and similarities, we establish that these are not the same.
- Most of the focus is given to UWB, which thanks to its characteristic features achieves high localization accuracies. The paper represents a generic tool for any indoor localization system designer, who might encounter similar problems.
- Future UWB research directions are described.

The following definitions remain valid throughout the paper and they are intended to help the reader understand the general concepts.

- Anchors are UWB nodes that form the infrastructure of the positioning system. They are typically installed in fixed locations.
- Tags are UWB nodes that can freely move in the deployment area.
- Self-calibration is intended as the automatic procedure that allows anchor nodes to precisely identify their own locations without additional interventions, e.g. manual measuring, or surveying. Typically, these nodes remain static after calibration.
- **Collaborative localization** is not only limited to anchor-tag communication, but it also involves peer-to-peer ranging, i.e. positioning is achieved without using fixed infrastructure nodes. In this case, the unlocalized nodes can be both tags and anchors.
- Self-positioning / self-localization is sometimes interchanged with the previous two terms. However, in this context we refer to it as the ability of a mobile node to calculate its own position, e.g., by passively listening to other communications happening. Moreover, this technique is performed at a given update frequency to track the position of the mobile nodes.

The rest of the paper is organized as follows. Section 2 presents related work. Section 3 analyzes the motivation of this survey more in detail. Section 4 outlines the most critical characteristics of UWB. Section 5 describes the problems and explain how they are typically tackled. Afterwards, section 6 reports the most significant research about the topic, followed by future work and the conclusion in section 7 and 8, respectively.

## 2 RELATED WORK

Previous localization surveys mostly investigated a wide range of metrics used to classify positioning methodologies. However, by looking at multiple aspects, some of the classifier are often overlooked. Even when collaboration is used to compare solutions, no further distinctions between self-calibration and collaborative localization are made.

#### 2.1 Categorization using networks

Although a clear categorization to distinguish between self-calibration and collaborative localization is not yet found in literature, work on collecting and analyzing articles on wireless RF-based positioning does exist [12–14, 19, 21, 29, 32, 39–41, 45, 48, 51, 58, 61, 91, 111, 114, 116]. The more general studies on wireless positioning with traditional performance comparison metrics can be found in [13, 48, 58, 111, 116]. Chowdhury *et al.* [13] mainly classify localization algorithms based on mobility and infrastructure. More specifically, on whether a positioning algorithm is based on anchor nodes Manuscript submitted to ACM

for which the initial location is given. Yassin *et al.* [111] provide an in-depth survey on cooperative localization in the sense of using multiple technologies, rather than the typical use of multiple unknown nodes collaborating within a single technology. This presents another gap within the classification of past survey papers and reinforces the need for a more thoroughgoing classification of the growing distinctions across cooperative localization systems. The use of UWB is discussed to a higher degree than other survey papers on indoor localization in [12, 51, 91].

Other studies, while maintaining traditional comparison metrics, have different main classifications. Surveys of wireless localization systems, which are based on mobile phones and mobile phone infrastructure are provided by [21, 61]. Maghdid *et al.* [61] present the main technical implementation challenges in smartphone localization technologies and solutions. The survey paper covers both the outdoor as well as indoor localization efforts, in addition to focus on seamless transition between outdoors and indoors.

The authors of [14, 19, 41] provide surveys with the focus on the processing performance of localization algorithms. Surveyed works by Coluccia *et al.* [14] are reviewed with three main comparison criteria: cooperation, mobility, and advances in signal processing. The cooperative localization methodologies are further subdivided as centralized or distributed.

The mobility of anchor nodes and unknown nodes are explicitly considered in [12–14, 39]. Chelouah *et al.* [12] use traditional performance metrics as the comparison criteria, with a focus on the mobility of sensor nodes in wireless sensor networks. A cursory classification of cooperative localization is one of the main comparison metrics in this paper. The classification excludes anchor-based localization methods from cooperative methods and limits the former to GPS-enabled devices.

Some comparison metrics for wireless positioning systems are defined by specific use-cases such as underwater localization [29] and emergency responds [32]. Ferreira *et al.* [32] provide the comparison of positioning systems by measurement type and localization methodology, with a particular focus on the requirements of emergency responders as comparison metrics.

#### 2.2 Categorization using algorithms

The authors of [45, 51, 114] compare localization systems by the algorithms and methodologies used. Laoudias *et al.* [51] classify wireless localization systems based on measurement type, enabling algorithms, with a focus cellular networks, data fusion, simultaneous localization and mapping (SLAM), and user mobility. The discussion on cooperative localization is limited. Jang *et al.* [45] compare localization works based on traditional metrics and the algorithms used. The three main distinctions made are triangulation, fingerprinting with offline map creation and fingerprinting without offline map creation. More specifically, the focus lies on SLAM algorithms, distance-based position inferences, and crowdsourcing methods. Zafari *et al.* [114] provide a technology independent distinction between positioning principles across the state-of-the-art. The authors make a distinction in the aspect of cooperative localization, depending on the perspective of either the infrastructure or the device to be localized. These are respectively defined as Monitor Based Localization (MBL) and Device Based Localization (DBL). The authors provide a thorough coverage of UWB-based localization systems.

Several works have compared positioning systems based on whether offline training is required [41, 91]. Shit *et al.* [91] contain a wide variety of characteristics to compare each discussed localization method. There is a particular focus on localization methodologies that require offline training and those which do not. The classification itself is thorough in comparison to other works. The limitation of this survey paper lies in the restricted variation in surveyed technologies. UWB is not specifically mentioned.

Surveys on collaborative localization in wireless sensor networks can be found as well. Probably, the most comprehensive work on cooperative localization has been done in [75]. Authors surveyed numerous aspects such as type of measurements, calibration, synchronization, localization algorithms and briefly on connectivity issues. In [8], authors discuss theoretical limits, algorithms, and practical challenges for collaborative localization. Both range and angle-based techniques are surveyed and explained. Attention is also given to less popular topics such as unknown noise variance, NLOS, multipath propagation and synchronization. Another survey focuses more on the algorithmic aspects of positioning in collaborative networks [56]. Authors review popular algorithms based on maximum-likelihood estimation, convex relaxation and optimization and message passing. The preferred measurement type is Received Signal Strength (RSS), as it represents a good candidate for low cost applications [65].

#### 2.3 Overview

Out the aforementioned 23 related works, 14 survey papers discuss cooperative localization, 10 papers explicitly classify by it and only [114] and [14] provide a further subdivision of cooperative localization. Similarly, positioning systems based on UWB are discussed in 12 of the reviewed survey papers.

We have shown that numerous surveys exist, but they focus mainly on non-UWB RF technologies. Moreover, none of the above studies explains nor highlights differences between self-calibration and collaborative localization. Even when these techniques are mentioned, they are not exhaustively analyzed, leaving space for a more detailed survey on these important topics for indoor positioning systems.

## **3 MOTIVATION**

As outlined in the introduction, UWB is often chosen for its high accuracy localization capabilities. Many articles show that position accuracy of 20 cm or less is possible [11, 49, 62]. Moreover, the large amount of scientific literature produced several comparative studies, which mainly focus on the UWB standard, its positioning techniques and its applications [89]. In some cases, a short paragraph is dedicated to less ordinary issues such as cooperative localization [23]. However, this topic contains various aspects, which require extensive study. In fact, there exist several studies about cooperation and self-calibration for indoor localization systems, as shown in Section 2. Thus, we want to collect, analyze and report what has been done for the specific problem of collaborative localization in indoor localization networks and focus on UWB cases or understand if the principles are applicable to UWB, when it is not explicitly mentioned. To this end, surveys which are not specifically based on the UWB technology are also included. In parallel, the same is done for self-calibration as this is closely related to cooperation but in a specific positioning sense (to localize the anchor nodes) to improve and optimize the installation procedures , by minimizing time and costs. Moreover, we distinguish between self-calibration and collaborative localization concepts, which are sometimes misused or interchanged. In fact, calibrating the infrastructure nodes requires cooperation among nodes in the network but since it involves only fixed nodes, it is more correct to refer to it as self-calibration procedure. Section 5 clarifies what is meant with these two concepts and introduces the classification, which we adopted in the paper. However, before going into details, we introduce general concepts of the UWB standard to help the reader better understand the underlying technology and align the terminology to avoid ambiguities. We keep the following section brief as it is not the scope of this paper to dig into UWB technology itself as existing works have covered this thoroughly [27].

# 4 UWB STANDARD OVERVIEW

UWB uses extremely short pulses in time, which translate, as the name suggests, in large bandwidth, i.e.  $\geq$  500 MHz. The UWB PHY works with two different modulation schemes, namely single pulse modulation and burst position modulation (BPM) / binary phase-shift keying (BPSK). The UWB frame is then modulated in two separate ways, which have different robustness. This is a specific aspect of this technology and could be exploited to improve localization algorithms. In Section 6.3, we show a way in which this characteristic can be used to achieve better performance. At the moment, the IEEE 802.15.4a is the reference standard for UWB communication. It allows UWB to work in the 3.1-10.6 GHz frequency band [70]. Since it operates in unlicensed bands, it is crucial to guarantee that UWB communication does not interfere with any other existing technologies such as WiFi in the 5 GHz band. Thus, Federal Communications Commission (FCC) establishes strict rules to ensure that UWB emissions remain small, i.e., low power spectral density (PSD) [98]. The Equivalent Isotropically Radiated Power (EIRP) limit is -41 dBm/MHz. This is an important aspect to consider when deploying UWB systems. Having low PSD has several implications, e.g., it could limit the maximum range the signal reaches still having high signal quality for localization purposes. However, UWB PHY defines the possibility to select among a large set of settings, which in turn will affect the performance in terms of accuracy, maximum range, and energy efficiency. These settings are:

- Frequency channel
- Pulse Repetition Frequency (PRF)
- Preamble length
- Bitrate

These are not the only things that can be changed but are the ones, together with the transmit power, that will influence UWB the most. This aspect is also discussed in Section 7.2 with regards to self-calibration and collaborative localization. Assuming that the best settings have been selected, which change from application to application, then UWB can be trusted in estimating distances with cm-level accuracy. By sending repetitions of short pulses, UWB can accurately timestamp incoming messages and more importantly distinguish peaks of the signals that belong to secondary paths. This is crucial in indoor environment where multipath propagation is the norm. However, a single distance is (almost always) useless to pinpoint the location of an asset regardless the environment. If 3D localization is the goal, then at least four separate distances are needed. An example of simple infrastructure with four anchors and one mobile tag is shown in Figure 2. The anchors are the devices that normally do not change their positions during the whole localization process, whilst the tag is the mobile asset that must be localized. In general, several techniques exist to do positioning in RF-based systems. These can be classified as shown in Figure 1. Scene analysis, proximity, and triangulation are



Fig. 1. Generic classification of approaches for localization systems. We focus on time-based solutions as specific case of range-based techniques [58].

three main categories that can be used to distinguish a localization algorithm as outlined in [58]. As UWB is often used in range-based systems, we focus on the triangulation category. This technique is in turn divided in angulation and lateration techniques. When the distances between two or more nodes are available, we differentiate based on how the distances are calculated and afterwards used. It can be done by looking at specific signal property, e.g., RSSI values [115] or by calculating the time that the signal takes to travel between the transmitter and receiver, known as Time of Flight (ToF) [37]. The former technique is typical of WiFi solutions, in which the location of the mobile asset is determined depending on the signal power levels of multiple access points [120]. On the contrary, thanks to the above-mentioned high timing resolution, UWB is often used in time-based distance estimation [88]. Moreover, UWB can either base the distance estimation on the arrival time at the receiver side (TOA) [35] or on the time difference of arrival among multiple receivers (TDOA). One of the most used distance measurement technique is two-way ranging (TWR). In TWR, tag and anchors are communicating in both directions. This exchange of messages can happen once, and this is called single sided TWR or it can happen twice, and, in this case, we talk of symmetrical double sided TWR (SDS-TWR) [83]. By accurately time-stamping the received messages (TOA) and calculating the ToF of each exchange, the distance between the devices can be estimated, assuming that the propagation speed is known. With this technique the range can be computed by the mobile nodes themselves and more importantly the anchors do not have to be synchronized. However, TWR requires more energy since all nodes are transmitting and receiving and more messages need to be exchanged. A solution to these two problems could come from adopting a TDOA approach. Hence, the scheme uses a one-way communication to achieve accurate localization. The mobile tag sends out a message that is then received by all anchors in range. These anchors must be accurately synchronized to be able to compute the difference in arrival times. Alternatively, the anchors can send out messages at specific non overlapping moments and the tag can compute the TDOA and therefore its own position. This technique has the benefit of reducing packet overhead, but time synchronization is crucial since a 1 ns error translates in 30 cm inaccuracy. In addition, TODA also results in lower latency compared to TWR, since a single measurement is received by multiple anchors. In TWR, it is not possible to perform measurements with more than one anchor at the time for a single transceiver. When the system exploits the direction of the propagating signal instead, we denote this as an angle-based solution. Typically, UWB works with the so-called Angle of arrival (AoA) algorithms [53, 68, 95, 108]. Multiple antennas are used to compare the estimation of the angles with the signal amplitude or the carrier phase, at which signals are received [1].

These are the general concepts used by UWB localization systems to track and localize mobile nodes. Although it seems straightforward to implement, there are many aspects to consider, especially when working in indoor environments. These aspects relate to signal propagation when subjected to obstructions, multipath and NLOS conditions. Despite the typical caveats of indoor localization, the characteristics of UWB provide opportunities to design solutions for aforementioned problems.

## 5 CLASSIFICATION CRITERIA

This section is intended to outline the classification criteria with which we survey the research on self-calibration and collaborative localization. UWB localization can be a simple task when a single tag is in LOS of 4 separate anchor nodes, as shown in Figure 2. However, sometimes anchors are physically obstructed by walls or metal structures creating NLOS conditions and reflections (Figure 3a). Furthermore, more generic scenarios consider multiple mobile nodes to be localized, as shown in Figure 3b. With the lack of sufficient anchors in range, a mobile tag's remaining option could be to cooperate with peer nodes to continue its localization efforts. This is a technique to which we refer as collaborative localization. Similarly, suppose there is a need to install additional anchors, e.g., to improve coverage in a specific area, Manuscript submitted to ACM

Ridolfi, et al.



Fig. 2. Typical UWB ranging accuracy (Actual – Estimated) is in the order of 10 cm. By combining multiple ranges from different anchors, the mobile tag can be positioned.



conditions on the direct path between tag and an-

(b) Multiple mobile tags in a densely crowded environment. For certain tags, the number of anchors within communication range are insufficient.

Fig. 3. Typical indoor problematic scenarios.

or to install a whole new system, it should be possible to do so in an automatic and reliable way. Placing anchors without having to manually measure their locations is called a self-calibration procedure.

We analyze differences and highlight similarities among different solutions for both self-calibration and collaborative localization, based on multiple criteria. In Figure 4, we present our classification approach. Starting from the general problem of having to position nodes in a network, we assume that the nodes are ranging with each other. If these nodes are all anchors, then we distinguish between the hypothesis of already knowing their location or not. We refer to the Manuscript submitted to ACM



9

Fig. 4. Although the main distinction is between self-calibration and collaborative localization, we identify extra features that have an impact when designing such a system.

first case as regular anchor-based localization, while we define the problem of having anchors ranging and not knowing their coordinates, as **self-calibration**. On the other hand, if the nodes that are ranging are not only fixed anchors, but also mobile tags and we want to localize them, then we classify these as **collaborative localization** problems. Moreover, independently from the type of problem, we classify different solutions based on other aspects as well. First, whether the position is computed in a distributed or centralized manner. Second, we differentiate on how positioning information is propagated throughout the network. We call it sequential approach when, once a node has determined its own coordinates, it will contribute to pinpoint other unlocalized nodes. The opposite happens in the concurrent case, i.e. localized nodes are not used in the computation of the position of other nodes [8]. Finally, we report the type of algorithm that is used to solve the problem. The idea is to give the reader a full understanding of the state-of-the-art techniques that are used now to tackle the self-calibration and collaborative localization problems. In summary, both of the following approaches would be possible: (i) deploy more anchors and let them calibrate themselves using other nodes in the network (self-calibration) or (ii) let some of the mobile tags act as anchors (collaborative localization).

Our goal is to unify as much as possible literature to avoid confusion and misuse of terms. However, it is important to keep in mind that sometimes terms have different legitimate meanings and interpretations. For example, in [121] Manuscript submitted to ACM



Fig. 5. Peers (tag or anchors) can talk to each other (collaboration) and additional infrastructure nodes, i.e. anchors, can be placed (calibration needed)

the authors refer to collaborative localization as the integration of different types of measurements, i.e. TOA- and RSSI-based.

In the next paragraphs, we explain in detail what is meant with each of the categories we have just presented. In Figure 5, we summarize the challenging conditions on which the systems operate. This image highlights a situation where few anchors are supposed to be used to localize many mobile tags. To manage such a scenario, more anchors can be installed to increase the coverage of the localization system (additional anchors) or alternatively, mobile nodes could be allowed to communicate to each other (peer-to-peer communication).

## 5.1 Distributed vs Centralized

Given the number of anchors and mobile nodes, the network can be organized in a distributed or centralized way. Typically, centralized networks require a backbone, which is used to deliver and collect all the information to a central point, e.g. to the localization server or engine. In contrast, in distributed networks all nodes participate in the position calculation [77]. Although distributed approaches are intrinsically more suited for nodes collaboration, centralized schemes are often used, e.g. when more computational power is required. Other aspects such as latency and overhead must be considered as well. Examples of network implementations for UWB systems can be found in [60, 82]. Both use a central entity and a separate backbone network to compute and deliver information to the UWB nodes, e.g. their time slots to transmit UWB packets. These are example of centralized schemes in which potentially latency and protocol overhead could limit the amount of node in the network itself.

#### 5.2 Sequential vs Concurrent

This distinction involves unlocalized nodes, either anchors or tags. In sequential approaches, localized nodes act as anchors for unlocalized ones. In this scenario, even mobile tags, once their coordinates are known, are used to compute other nodes' positions, which could be both another tag and an anchor [8]. Conversely, concurrent behaviour means that once an unlocalized node has been localized, it will not be included to compute the position of other unlocalized nodes. The benefit of the concurrent approach is that errors will not be propagated to other nodes their positioning calculations. In sequential localization [8], a node has typically more chances to be localized as there are more available nodes that are considered for the position computation.

## 5.3 Type of algorithm

The last aspect we want to consider is the type of algorithm used to calculate the position of the unlocalized nodes, as classified in Figure 1. This is done to give the reader the opportunity to compare different solutions and to understand the potential of each of them, e.g., RSS-based localization (triangulation) could be applied both to WiFi and UWB standards. Some examples of most used localization algorithms are reported in Table 1, which also reports the type and the basic principle for each solution. Nevertheless, advanced algorithms sometimes consist of hybrid solutions in which two or more techniques are simultaneously used to solve the localization problem.

Algorithm	Туре	Basic principle
Multilateration	Lateration	Estimating the location of the tag based on the time difference between signals arriving at different anchors [72]
Multidimensional scaling (MDS)	Scene analysis	An exploratory method that analyzes the dis- similarity of data for a certain set of objects [5]
Maximum likelihood estimation (MLE)	Scene analysis	A method that by maximizing the likelihood function, estimates the parameters of a proba- bility distribution [50]
Kalman and particle filters	Scene analysis	Algorithms that uses noisy observation over

time to estimate the state of an unknown vari-

ables [84]

Table 1. Frequently used algorithms to solve localization problems

## 5.4 Self-calibration vs collaborative localization

The most important and widely used solutions are presented in this section and the most relevant approaches for UWB (in which UWB characteristics may be exploited) are described more in depth. As already mentioned, this is the most controversial separation in literature. The difference between collaborative localization (involving both unlocalized tags and anchors) and self-calibration (exclusively among unlocalized anchors) is subtle but still very relevant. The main principle is the same: a node that does not know its position uses not only fixed anchors but also other localized nodes. Moreover, communications are not only between anchors and tags but also between peers, e.g. tag-tag or anchor-anchor. Whether these agnostic (not localized) nodes are both tags and anchors or exclusively anchors, defines the two problems, which are collaborative localization and self-calibration, respectively. We believe Manuscript submitted to ACM that these two cases must be differentiated because they also require different approaches. In an area wherein manual calibration is not suitable, whether an infrastructure needs to be expanded or is not present at all, the ability of the anchors to self-calibrate is crucial and must be as accurate as possible [71]. Moreover, all newly calibrated anchors must always support all tags, thus requiring sequential solutions to solve the calibration problem. In the collaborative localization approaches, all the fixed nodes , if present, are already in place with known coordinates and there is no direct need to add new ones, the mobile tags could help each other [20, 105]. A tag can also act as anchor to increase the accuracy of the system. However, Section 6 shows that the accuracy does not always increase for UWB systems, especially in critical scenarios, e.g., NLOS working conditions. When a mobile tag acts as anchor, the algorithm needs to be carefully designed to consider other related issues such as mobility of the node, reliability of its position and network synchronization.

Table 2 Summary	v of major studies on	self-calibration an	d collaborative	nositioning f	or indoor	localization systems
Table 2. Summa	y of major studies on	sen campiation an	u conaborative	positioning it	or muoor	iocalization systems

Solution	Nodes type	Topology	Type of algorithm	Sequential Concurrent	UWB specific	Exp. evaluated
Hamer et al. [38]	Anchors	Distributed	Triangulation	Sequential	Y	Y
Vashistha et al. [103]	Anchors	Centralized	Triangulation	Sequential	Υ	Y
Batstone et al. [3]	Anchors	Distributed	Scene analysis	Sequential	Y	Y
Rajendra et al. [80]	Anchors	Distributed	Triangulation	Sequential	Y <sup>1</sup>	Y
Shi et al. [90]	Anchors	Not specified	Scene analysis	Sequential	Y	Ν
De Preter et al. [78]	Anchors	Centralized	Scene analysis	Sequential	Y	Y
Yu et al. [113]	Anchors	Not specified	Triangulation	Sequential	Y	Y
Dardari et al. [20]	Both	Not specified	Scene analysis	Sequential	Y	Ν
Maissner et al. [63]	Both	Not specified	Triangulation and scene analysis	Concurrent	Y <sup>1</sup>	Ν
Cai et al. [9]	Both	Centralized	Scene analysis	Sequential	Y	Y
Van de velde <i>et al.</i> [101]	Both	Centralized	Scene analysis	Sequential	Y	Ν
Van de velde <i>et al.</i> [100]	Both	Not specified	Scene analysis	Concurrent	Y <sup>1</sup>	Ν
Wymeersch et al. [107]	Both	Distributed	Scene analysis	Concurrent	Y	Y
Xu et al. [109]	Anchors	Centralized	Scene analysis	Sequential	Ν	Y
Patwari et al. [76]	Both	Not specified	Triangulation and scene analysis	Sequential	Ν	Y
Moses et al. [69]	Anchors	Centralized	Triangulation and scene analysis	Sequential	Ν	Ν
Doherty et al. [28]	Both	Centralized	Triangulation	Concurrent	Ν	Ν
E. Larsson [52]	Both	Distributed	Triangulation and scene analysis	Sequential	Ν	Ν
Costa et al. [16]	Both	Distributed	Scene analysis	Sequential	Ν	Y
Savvides et al. [86]	Partial	Centralized	Triangulation	Sequential	Ν	Y
Čapkun et al. [10]	Anchors	Distributed	Triangulation	Sequential	Ν	Ν
Ihler et al. [43]	Partial	Distributed	Scene analysis	Sequential	Ν	Ν
Kim at al. [47]	Partial	Distributed	Triangulation	Sequential	Ν	Ν
Fox et al. [33]	Anchors	Distributed	Scene analysis	Sequential	Ν	Y
Mekonnen et al. [64]	Anchors	Centralized	Scene analysis	Sequential	Ν	Ν
Crocco et al. [17]	Anchors	Distributed	Scene analysis	Sequential	Ν	Ν
Priyantha et al. [79]	Anchors	Distributed	Triangulation	Sequential	Ν	Ν
Zhou et al. [118]	Anchors	Both	Scene analysis	Sequential	Ν	Ν
Jia et al. [46]	Both	Centralized	Scene analysis	Sequential	Ν	Ν
Li et al. [55]	Both	Distributed	Scene analysis	Sequential	Ν	Y
Vaghefi et al. [99]	Both	Centralized	Scene analysis	Sequential	Ν	Ν
Gholami et al. [34]	Both	Distributed	Scene analysis	Sequential	Ν	Ν
Yin et al. [112]	Both	Distributed	Triangulation and scene analysis	Sequential	Ν	Ν
Tomic et al. [96]	Both	Centralized	Triangulation and scene analysis	Sequential	Ν	Ν
Sathyan et al. [85]	Both	Not specified	Triangulation and scene analysis	Sequential	Ν	Y
Li et al. [54]	Both	Distributed	Scene analysis	Sequential	N	N

## 6 LITERATURE OVERVIEW

Table 2 lists an extensive overview of the major studies on both self-calibration and collaborative localization techniques [3, 9, 10, 16, 17, 20, 28, 33, 34, 38, 43, 46, 47, 52, 54, 55, 63, 64, 69, 76, 78–80, 85, 86, 90, 96, 99–101, 103, 107, 109, 112, 113, 118]. In the next paragraphs, we focus on those articles that already target UWB as main technology. Nevertheless, we extended the classification work by using two extra columns in Table 2, namely UWB specific and experimental evaluation. By doing so, the reader gets a more general understanding of each solution and should note that even when non-UWB specific, the concepts may also be applied to other technologies. This choice also reflects the idea of enabling future research directions by showing that good work has already been done but there is still space for enhancement in Section 7. To emphasize the main differences between self-calibration and collaborative localization, we describe the

<sup>&</sup>lt;sup>1</sup>UWB-specific features are used to improve accuracy and overall performance

two concepts in separate sections. An additional paragraph is dedicated to solutions that not only use UWB to validate their solutions but also design algorithms using UWB-specific features such as channel impulse response (CIR) or NLOS detection.

A wide variety of solutions (to the same problem) is reported in Table 2. These large differences among all papers make the performance comparison very challenging, as each implementation uses proprietary metrics. A solution to objectively compare the performance of multiple solution is outlined in [102]. Authors present the EVARILOS Benchmarking Platform. The idea is to enable automated evaluation and comparison of multiple solutions in different environments, by introducing a testbed-independent benchmarking platform, combined with multiple testbed-dependent plugins. However, such an objective evaluation method was not used for most of the scientific papers discussing the algorithms that we considered. Therefore, we report results mainly in terms of accuracy and we indicate, as clear as possible, the conditions in which each evaluation took place, e.g., in Table 3 we list the environment, the operating conditions and the size of the area for each study.

#### 6.1 Self-calibration with UWB

When fixed anchor nodes are deployed with their positions unknown to the system, there is a need to calibrate the anchors in the sense of determining their locations to localize a mobile asset in any indoor environment. Their positions can be absolute or relative to each other. Either way, the calibration procedure can be carried out manually with special equipment (laser meters, camera, etc.) or in an automated way. If no manual intervention is required or possible, we refer to self-calibration procedures. As shown in Figure 1, localization is generally achieved with multiple techniques and types of data. However, works on self-calibration with UWB as listed in Table 2 are based on range measurements. Thus, to compute the location of the unlocalized anchors, the system must first know the distances between anchors. Once these are collected on a central point or across all nodes, different algorithms can be applied to compute the position of the anchors. The self-calibration problem requires a sequential approach. In fact, positioning any other mobile tag can be achieved by using the newly localized anchors. Various self-calibration approaches using UWB are reported in seven of the studies in Table 2 [3, 38, 78, 80, 90, 103, 113]. All of them are designed and/or validated using UWB indoor localization solutions. The main assumption is that no infrastructure is deployed. Mobile tags use anchors that are being calibrated by means of self-calibrating anchors. Therefore, the first step of these systems is to (self) calibrate the fixed anchor nodes to create a reference system. However, there exist various approaches to solve the issue.

In [38], the algorithm to self-localize the anchors is a distributed and iterative approach in which three anchors are initialized to define the reference system using the messages exchanged among them. After this first phase, the position estimates are further refined using distributed gradient descent. Ultimately, each anchor communicates its position to all other anchors in the network. The authors do not exploit any UWB-specific characteristic that could improve the performance of their solution. However, they reported experimental results with a position Root Mean Square Error (RMSE) of 97 mm to self-localize 8 anchor nodes [38]. Moreover, the mobile nodes can calculate their positions using TDOA by passively listening to the messages exchanged among the anchors, assuming that the system is accurately synchronized. The adopted scheme is a distributed and sequential solution using range-based measurements.

The idea to adopt TDOA is also used in [103], wherein a sequential and centralized low-cost self-calibration scheme is designed. Assuming the nodes are synchronized, the authors let the anchors range with each other (using a predefined transmission scheme) and by further computing the Differential TDOA (DTDOA), any anchor can be calibrated and its coordinates used in the positioning of mobile nodes (sequential solution). This solution is experimentally evaluated with up to 4 anchors. The distances used between the anchors were 5, 7, 10 and 15 meters in a controlled office environment. Manuscript submitted to ACM

Compared to the TDOA solution presented in [38], the RMSE was significantly larger from 17 cm up to 31 cm when calibrating 4 anchors.

Another sequential but this time distributed UWB research that assumes unknown the spatial coordinates of the anchors, is presented in [3]. In this study, a novel RANSAC approach with outliers and missing data was constructed and experimentally evaluated with UWB drones. As for previous work, the authors tested their solution in a small room with MOCAP system and obtained accuracy of around 11 cm for 6 anchor nodes. However, they then tested the deployed the same anchors to office space with multiple rooms and despite a 49.61% of data missing, due to more challenging environment, they still achieved better results than what was reported in [103], i.e. a mean accuracy of 13 cm.

An easy-to-use strategy for anchor self-localization is the focus of [90]. The positions of the anchors are automatically derived by combining UWB ranging estimations and low-cost Inertial Measurements Unit (IMU) data. This is the first solution we describe that uses other type of sensors rather than only UWB distances to compute the position of the anchors. The applied sequential method is based on SLAM technique and utilizes Error-State Kalman Filter (ESKF) to fuse UWB and IMU readings, resulting in UWB anchor position estimates and six Degrees of Freedom (6DoF) tag pose estimates. Given the different nature of the algorithm with respect to [3, 38, 103], for the system to work properly, freely moving the tag in the space of interest is the only thing that is required. Such a theoretical model is validated with several simulated experiments and 5 anchors, which show a typical RMSE of less than 15 cm in each space component, which is in line with what other studies have obtained in similar conditions.

Instead of solely relying the ranging capabilities of UWB, in [78], the authors also look at exploiting and modelling the UWB bias. A range bias correction is applied to distance estimations used in the auto-calibration technique presented. Implementing a centralized MLE algorithm, the authors achieve accurate positioning of mobile nodes, with an average error of 9 cm, using the auto-calibrated anchors, thus using a sequential solution. In this analysis, the authors used 4 anchors to test their solution as other evaluation we presented did as well. They also outperform previous solutions in terms of accuracy. However, the algorithm uses a second technology, namely RTK-GPS, to refine the initial guess of the anchors' locations. This is of course still a limitation for indoor scenarios wherein GPS signal cannot propagate.

Similar experimental results as in [90] are found in [113]. A two-process algorithm called SELF-CAL is proposed. Authors first apply Markov State Transition Equation (MSTE) to calculate the state vectors of all coordinates of the anchors and then, by using an iterative trilateral localization method, they calculate all the coordinates step by step in a sequential way. The calibration errors are once again averaging 10 cm in this solution. In addition, the authors showed that using the newly calibrated anchors and moving tag, the maximum dynamic error reported is less than 50 cm and the averaged error around 30 cm when compared to real track.

Overall, self-calibration with UWB allows to use many different types of algorithms as shown in the analysis. All the solutions we presented are range-based, to prove one more time that UWB typically works by estimating distances (or angles) between devices. Another common aspect is the relatively small number of anchors that are used to validate the system in experimental evaluations. Only in two cases more than 4 anchors are used [3, 38]. This is an important aspect to keep in mind when analyzing the results in terms of calibration accuracy as adding more nodes, in much larger environments (more information in paragraph 6.4) could affect the performance of the algorithms. All surveyed works use sequential solution as expected since anchors are then used by other nodes in the network as reference points and the accuracies among different solutions are comparable, in the order of 20 cm localization errors.

#### 6.2 Collaborative localization with UWB

In collaborative localization, the key point and main difference with previous self-calibration concept, is that there is no requirement to have fixed anchor nodes with or without known locations. All nodes in the network are considered peers and are typically communicating with each other to exchange several information, such as the distances, to compute their positions. When the location is derived, nodes do not have necessarily to be used again to localize other devices, thus concurrent solutions are in this case possible. A distributed approach would be the more logical to solve the cooperation problems in wireless networks, however limited computational capabilities and power could be the bottleneck for distributed collaborative algorithms, especially for the mobile nodes. Thus, some of the solutions in Table 2 are designed with a centralized scheme [9, 101]. Once more, we focus on describing and comparing those solutions that have been designed and tested specifically for UWB. All the solutions that are presented in this paragraph use iterative optimization algorithms. At each iteration step, the locations of the nodes are refined based on the current states. What changes from one algorithm to the other is the role of these newly localized nodes. If they are then used in the calculation for other unlocalized nodes, then we refer to it as sequential solution. If not, we call it concurrent collaborative localization, as explained in Section 5.2.

The approach presented in [20] compares UWB situations in which collaborative (among mobile nodes) localization is either used or not used. In this circumstance the anchors are in place and their coordinates are known. Interestingly, if cooperation among tags is used, it does not mean necessarily improved accuracy. As a matter of fact, two things need to be considered when selecting mobile nodes as 'anchor': (i) whether LOS conditions were present or not and (ii) the geometry of the devices. Moreover, the used iterative LS algorithm showed that the geometric configuration of the nodes may have a stronger impact than the accuracy of the distances between nodes on the overall localization performance.

Authors in [9] suggest a strategy to improve localization accuracy in typical indoor NLOS environments. In line with findings of [20], they argue that normally, this kind of scenarios always leads to insufficient LOS ranging to perform multilateration and using NLOS measurements means large accuracy errors. To solve this, the authors avoid using NLOS rangings and use LOS between mobile nodes instead (in a sequential way). They define a non-linear optimization problem and use Simulated Annealing (SA) to solve it. Their approach is tested with commercial UWB hardware and errors in the order of 10 cm are reported when only LOS measurements are used (50% improvement from using NLOS estimations as well). The tests are conducted with 4 anchors and 2 unlocalized tags in a 10m x 5m environment.

In [101], message passing algorithms are compared to investigate the benefits of cooperative localization in NLOS conditions. More precisely, two algorithms called SPAWN (based on belief propagation) and SLEEP (expectation propagation) are described. To enable a fair comparison, the authors modified SLEEP to work with reduced network traffic and to include the reference nodes (ASLEEP version). Authors simulate a scenario with 100 users and 13 anchors using Monte Carlo simulations. The range of each anchor is fixed at 20 m in a total area of 100m x 100m. Although ASLEEP is less complex and requires less traffic than SPAWN, simulations show that SPAWN performs slightly better, in particular when increasing the percentage of NLOS messages. In LOS-only scenario, the difference with SLEEP is small and they both score around 50 cm error. These algorithms are both sequential, centralized and use range-based information.

We found another example of the use of SPAWN for collaborative localization. Authors in [107] present a fully distributed cooperative localization algorithm (SPAWN), which is also validated by means of realistic ranging models. Moreover, the solution is simulated with 13 anchors and 100 tags, same as in [101] but smaller area, i.e. 100m<sup>2</sup>. Dealing



(a) CIR data from LOS conditions between two devices. The first path is well above other secondary peaks. Thus, timestamp can be accurately estimated.



(b) CIR data from NLOS conditions between two devices. The overall response is attenuated with respect to the above LOS case, even though PHY settings are the same. CIR analysis shows that secondary path come in stronger than the direct path, suggesting obstacles between transmitted and receiver.

Fig. 6. Channel Impulse Responses (CIRs) from DW1000 transceivers in different conditions, show different behaviour.

with mixed LOS/NLOS conditions is also investigated to evaluate the performance of the algorithm. However, the numerical results and the meter accuracy they achieved, leaves the question open to advanced investigation as results are far from what shown possible in [101].

We have shown that a great number of solutions that primarily focus on UWB already exists [3, 9, 20, 38, 63, 78, 80, 90, 100, 101, 103, 107, 113]. However, none of them in fact makes use of UWB-specific features (see Section 4) to boost the performance. Fortunately, there are some exceptions to this, and they are explained in the following passage.

#### 6.3 Exploiting UWB characteristics

In Section 4 we outlined some of the most important characteristics of UWB. Without a doubt the main advantage of this technology is the ability to properly identify secondary paths in the received signal. Multipath propagation is Manuscript submitted to ACM

commonly caused by reflections and scattering from obstacles in the vicinity of the transmitters and receivers. In Figure 6, a comparison between LOS and NLOS scenarios is depicted. Both plots report the CIR with a significant difference. In Figure 6a the first path is clearly identified, and its amplitude is the highest, thus making it easier for the receiver to timestamp the incoming packet, e.g., by using Leading Edge Detection (LDE) algorithms [73]. On the contrary, the overall CIR in Figure 6b is significantly more attenuated and the secondary peaks higher than the first one, which represents the direct path between transmitter and receiver. This second CIR is a clear indication that communication happened in NLOS conditions. Wrong peak detection could explain larger ranging errors in heavy NLOS conditions. However, predicting and correcting these errors, for example through Convolutional Neural Networks (CNNs), may boost the algorithm performance even in complex and larger solutions.

Moreover, UWB is particularly robust because of the different modulation schemes that it adopts. For both selfcalibration and collaborative localization, it is important to know how many other nodes are available in a network, i.e., the so-called discovery phase. However, operating conditions are not always known nor static. Thus, the system might not know the best PHY setting to use to reach as many nodes as possible. Thanks to single pulse modulation, by receiving only the synchronization header (SHR), the full CIR can be recorded and all information attached to it can be extracted [36], e.g., LOS/NLOS, presence of obstacles, power associated with the response and so on. From this, the system can derive the best set of settings to use not only for the whole system but for each device-to-device pair, since each combination might be operating in different conditions. These and other aspects of unique UWB possibilities are described in Section 7.

Although attempts to improve accuracy of self-calibration and collaborative localization by exploiting specific features of UWB, these are typically limited to discard ranges that happened in NLOS conditions. In fact, in scientific literature, examples of this kind of approach can be found and the next two paragraphs are dedicated to these solutions.

*6.3.1* Self-calibration. Differently from what we showed in Section 6.1, the authors in [80] uses extra features to improve the performance of the self-calibration. They propose a real-time, infrastructure free self-calibration scheme, which considers a highly connected network. The proposed method exploits UWB specific characteristics to filter NLOS communication links, simply by comparing total received power and the first path power. A theoretical ranging update rate of 50 Hz is proposed for a network consisting of 4 nodes. The authors do not provide a comment on the scalability of this method, which is based on a highly connected network. The proposed method reports experimental positioning errors less than 10 cm and a positioning update rate of 10 Hz. Even in more challenging NLOS environments, by looking at signal properties, the authors achieved comparable accuracy as in [3, 38, 78, 113], which are evaluated in LOS only.

*6.3.2* Collaborative localization. A special solution for the collaborative localization problem is presented in [63]. Authors designed an algorithm that considers a characteristic of UWB signals: their channel impulse response (CIR), which among others, is typically used to decide whether the communication between two devices has happened in LOS conditions or not. Moreover, reflections can be easily isolated and filtered out. From this, knowing the geometry of the area they are covering, they use 'virtual anchors' (VAs) to locate the mobile nodes. VAs are virtual nodes, generated from signal reflections of known anchor nodes. Thus, authors tested their concurrent TOA-based solution with Kalman and Particle filters to localize mobile nodes and obtained accuracy of about 45 cm in typical indoor pedestrian scenario. Having virtual anchors, thus reducing costs of installing physical ones, also means lower accuracy if we compare these results to the previous ones in Section 6.2

The concept of VAs is also studied in [100], which introduces a low-cost two step algorithm called CUPID. This is also a case of concurrent TOA-based algorithm. It is claimed that even with a single anchor and exploiting collaborative Manuscript submitted to ACM

localization among nodes, it is possible to perform accurate localization. Additionally, this is viable by integrating the floor plan, which produces a set of VAs. In the first step, exploiting collaboration among users, their relative coordinates are estimated and then (second phase) transformed to absolute coordinates. To this purpose, multipath components (MPCs), which are easy to extract thanks to the nature of UWB signals, are also considered for the absolute coordinate transformation. Simulations show good results in LOS condition, i.e. localization error < 30 cm. On the other hand, NLOS conditions cause a significant degradation of performance.

## 6.4 Reference environment

Table 2 includes a column to differentiate between algorithms that are experimentally evaluated and those that are not. Nearly half of the reported papers have done tests in real life conditions. In particular eight of the UWB specific works have done so [3, 9, 38, 78, 80, 103, 107, 113]. They all have in common the evaluation metric, i.e. the accuracy of UWB ranging and positioning but they differ in an essential aspect: the environment in which tests took place. As we outlined in Section 4, the surroundings are very important to assess the performance of UWB. Table 3 shows where these eight solutions where tested. Moreover, it highlights the operating conditions and the size of the area. Almost all of the

Solution	Environment	Conditions	Size
Hamer <i>et al.</i> [38]	Office/Lab	LOS	6x7 m <sup>2</sup>
Vashistha <i>et al.</i> [103]	Office/Lab	LOS	15x15 m <sup>2</sup>
Batstone et al. [3]	Office/Lab	LOS	3x2 m <sup>2</sup>
Rajendra <i>et al.</i> [80]	Building hallway	LOS/NLOS	$22x20 \text{ m}^2$
De Preter et al. [78]	Open space (outdoor)	LOS	17x6 m <sup>2</sup>
Yu et al. [113]	Office/lab	LOS	$10x8 m^2$
Cai <i>et al.</i> [9]	Office	LOS/NLOS	$10x5 \text{ m}^2$
Wymeersch et al. [107]	Labs and hangar	LOS/NLOS	$12x20 \text{ m}^2$

Table 3. Type of environment in which UWB is evaluated

solutions are tested in an office space environment, in a very controlled scenario and reasonably small. Especially, those testing only LOS conditions [3, 38, 78, 103, 113] are on average performed in smaller spaces. Only four times, more challenging conditions (NLOS) were part of the tests. Still, the size of the environment is limited to around 10x15 m<sup>2</sup> on average, which is relatively small compared to other scenarios of interests such as warehouses and entire buildings.

# 7 FUTURE RESEARCH DIRECTIONS

Solid work on self-calibration and cooperation among mobile tags has been done as shown in Table 2. As we want to focus on a specific technology, i.e. UWB, we believe that much more could be achieved. Multiple research questions are still unanswered. Among others, we identify the following: (i) exploiting UWB ranging error estimation and correction, (ii) adapting the UWB PHY layer, (iii) compatibility, (iv) multiple evaluation metrics, (v) end-to-end system aspects and (vi) applications. In addition, the market growth, and the consequent opening to new sectors (UWB in smartphones), also brings new tasks especially related to self-calibration and collaborative localization, e.g. smartphone communicating to Manuscript submitted to ACM

each other without infrastructure. The following paragraphs explain challenges and open questions on self-calibration and cooperation in UWB systems in more detail.

## 7.1 Exploiting UWB ranging error estimation and correction

UWB is uniquely suited for high accuracy ranging due to the use of short time pulses. An example of exploitation of these unique UWB features, can be found in [6], wherein the authors propose two methods to reduce localization error without prior knowledge about the radio environment. The authors propose the use of CNNs to (i) perform NLOS channel classification and (ii) to create ranging error regression models. Both methods can be used as additional input for self-calibration or collaborative localization, providing not only a range estimate but also an indication of the reliability off the estimation. Interestingly, it is shown how using raw CIR data outperforms approaches based on derived input signal features to detect NLOS conditions. Applying this to self-calibration or collaborative localization, or be used as a divident could significantly improve the performance of the two techniques we investigated in our analysis. Even more generally, other causes of errors could also be considered, providing much more information regarding the reliability of individual ranging measurements. Table 4 summarizes the most common sources of inaccuracies, many of which are not yet considered in self-calibration approaches. These causes of errors range from PHY-based aspects to network and application ones, highlighting the importance of a 'full-stack' approach when designing UWB systems. In literature

Table 4. Most common sources of errors in UWB localization systems

Cause of error	Possible mitigation approach
Pulse Distortion	Antenna and TX power calibration
Clock drift	Double sided TWR algorithm
Range Bias	Model to correct bias, e.g., on RX power
NLOS conditions	Including CIR in the range calculation
Latency	TDOA algorithm
Scalability	Scheduled access MAC protocols
Large DOP	Optimization of anchors placements

[36, 66, 67, 92, 104], most papers focus on reducing the error of at most one of these causes of errors, while the majority of the solutions we presented in Section 6 do not correct the bias or error at all.

#### 7.2 Adapting the UWB PHY layer

The possibility of using different physical settings for improving communication reliability has already been studied in [36]. The authors define a mechanism that changes settings such as channel, pulse repetition frequency, preamble length and data rate at runtime. Moreover, they also use the estimated CIR to measure the link quality of the environment and based on this, change, and adapt the PHY parameters. Experiments show increased communication robustness and energy efficiency, proving that this is a crucial aspect for IEEE 802.15.4a systems. Authors in [36] have shown that only the first part of the UWB frame is needed to record this information and adapt the PHY layer settings. This is important to claim that energy consumption can be kept low as reception of the full frame is not needed for certain Manuscript submitted to ACM

phases of the self-calibration and collaborative localization processes, i.e., device discovery. However, all works related to self-calibration assume that the UWB device operates using a single RF configuration, and thus do not consider the impact of different UWB settings. For self-calibration, a possibility is to use higher TX power to reach a larger number of anchors and perform a more reliable calibration [25]. Nevertheless, the impact on accuracy and the possibility that the system would not be operating within legal limits have to be considered as well. In the user manual of the popular DW1000 UWB chip [24], there is already a suggested distinction between settings that should be used in LOS and NLOS. Table 5 reports these parameters. Preamble length, pulse repetition frequency and noise threshold multiplier differ

Table 5. Decawave suggestions based on the operational conditions, LOS or NLOS [24]

Parameter	LOS	NLOS
Preamble length	≥ 128	≥ 1024
Pulse repetition frequency	16 or 64 MHz	16 MHz
Noise threshold multiplier	13	12

between the LOS and NLOS case to basically exploits reflections and increase the sensitivity of the direct path signal in the NLOS cases. At the same times, these settings will increase the chance to erroneously trigger the reading of the incoming packet, which should be then corrected by the system. A self-localization system could continuously switch between different PHY settings to exploit the different range and accuracy trade-offs each of these settings offer. As such, the use of multiple UWB PHY settings to (i) increase the number of anchors within range and (ii) better correct environmental conditions will be beneficial to consider during self-calibration, making the measurements more reliable.

### 7.3 Compatibility

Currently, different commercial and academic UWB systems are designed independently of each other, using a variety of UWB settings, ranging approaches, MAC protocols and frame formats. To avoid a wild growth of incompatible ecosystems, there is a strong need for increased interoperability of future UWB systems. For this reason, several new industry alliances (UWB alliance [2], Fine Ranging Consortium (FiRa) [15], etc.) have recently been created with the aim of defining interoperability specifications for different application domains. One noteworthy initiative is the IEEE 802.15.4z standardization task force [87]. Besides defining future UWB PHY layers, the group is currently also designing new Information Elements (IE's) that can be added to existing IEEE 802.15.4 UWB frames, allowing negotiation of ranging methods and PHY settings between devices. Similar information elements could be defined for the purpose of negotiating the details of self-calibration and collaborative localization approaches, or to indicate a switch from localization to self-calibration.

#### 7.4 Multiple evaluation metrics

The research papers [3, 9, 20, 38, 63, 78, 80, 90, 100, 101, 103, 107, 113] typically focus on improving localization accuracy. However, they do not consider other essential evaluation criteria that are relevant for realistic deployments, such as suitability for constrained devices, energy consumption, latency, etc. As such, there is a need to investigate existing trade-offs between performance and complexity. For example, researchers can look for drop-off points after which additional collected measurements result in diminishing accuracy returns or evaluate the memory and computational Manuscript submitted to ACM overhead of distributed solutions on embedded devices. In addition, studies on combining different positioning methods such as ToF, RSS and sensor fusion are also relevant to evaluate the impact of complex algorithms on accuracy, energy consumption and overhead.

#### 7.5 End-to-end system aspects

Most scientific papers currently assume self-calibration to be an independent aspect of the whole system. In practice, self-calibration is merely an intermediate phase before (or while) the system is operational. As such, there is a need for self-calibration solutions that do not operate independently, while assuming no tags are being localized, but instead continuously verify and recalculate the location of the anchor nodes during the operational lifetime. To this end, messages that are being transmitted for localization or communication can be reused. As an example, backtracking particle filters have been proposed that are able to correct past location predictions of mobile tags based on more recent location estimates [4, 42]. Such backtracking algorithms [97], that aim to minimize the error of range estimates considering recent and past locations, could be modified to also correct the most likely position of the anchor nodes. As far as the authors are aware, such research about the optimal combination of operational aspects and auto-calibration phases is not yet explored.

#### 7.6 Applications

Table 3 lists different environments in which UWB systems are deployed and tested in the considered scientific papers. Most of these scenarios are limited to small size offices and laboratories. However, many other applications are gaining interests. An example is automation for warehouses with drones autonomously flying in large spaces [59, 81]. These industrial environments are typically characterized by the presence of metals [22], which create multipath, reflections, diffraction, and obstruction. As shown in some works [9, 80, 107] UWB can pinpoint the location of the mobile nodes with good accuracy. However, more studies are needed to identify the feasibility of such a system in a large environment, having to cope with many technical constraints, e.g., size of the areas, obstacles, and speed of mobile nodes. Moreover, the integration of UWB in the iPhone 11 [44], which will probably be followed by other phone manufacturers, also indicates that UWB can be used for diverse application in the future.

Alongside industry 4.0 related applications, the new upcoming standard IEEE 802.15.4z [87] defines several new application domains. The security aspects of the new standard make the UWB technology suitable for keyless entry applications in automotive or building automation [74, 110]. For example, the mobile phone can act as key to open the door of the car by proximity detection. Similar concepts can apply to secure mobile payments. Temperature sensing in greenhouses is another concrete case in which collaborative localization among nodes could improve efficiency and performance of the operations [93]. Having dozens or hundreds of battery powered sensors collaborating to derive their own positions, could save time and money as the devices are sensing information that most of the time is position related, e.g., a specific area of the greenhouse gets too warm during the day and certain plants must be placed somewhere else.

Interestingly, UWB has proven to be a flexible standard that can be used in many situations. An example is also the social distancing application during the SARS-CoV-2. Solutions to provide accurate and reliable distances estimations to keep people far enough from each other have emerged. This is an interesting example of how peer-to-peer communication can take place with UWB, i.e., collaboration between nodes.

What works best in LOS conditions may fail when metal obstacles are present but still work when plants are obstructing Manuscript submitted to ACM

the LOS between devices. Hence, it is crucial to exploit UWB related features to ensure that an optimal configuration is used in each type of application.

## 8 CONCLUSION

With many research articles on self-calibration and cooperation for indoor localization systems with focus on UWB, we decided to summarize the most researched approaches to give the readers the tools to better investigate and understand different opportunities. Features and possibilities of both self-calibration and collaborative localization are presented by separating the two concepts for a better understanding. Mainly the distinction between self-calibration and collaborative localization for UWB is resolved. Self-calibration indicates the specific procedure of acquiring/finding the coordinates of the infrastructure nodes, i.e. anchors. In contrast, in collaborative localization, the unlocalized mobile tags communicate with each other to estimate their own positions, i.e. without necessarily using a fixed infrastructure.

Although most of the reported solutions can be applied to a wide range of positioning technologies, we targeted those that used UWB to design and demonstrate the different application possibilities. We focused on UWB as its unique features represent a great potential for a wide range of indoor localization systems. Most scientific papers just treat self-calibration and collaboration problems as pure localization dilemma, and UWB-specific features, e.g. high timing resolution to mitigate multipath, are not exploited at all. As such, we listed a range of future research directions and strategies to better utilize UWB specific characteristics of the indoor localization problems. These include the use of better error mitigation models, the use of dynamic UWB PHY settings, providing UWB specific information elements for interoperability during self-calibration, considering multiple evaluation criteria while evaluating self-localization approaches, considering the interaction between localization and self-calibration phases and considering new application domains such as automotive. These directions will inspire future research paths for this promising domain.

#### 9 ACKNOWLEDGEMENTS

This work has partially been funded by the following projects: VLAIO proeftuinproject SmartConnectivity, imec.icon InWareDrones and EOS project MUSE-WINET.

#### REFERENCES

- [1] Abdulrahman Alarifi, AbdulMalik Al-Salman, Mansour Alsaleh, Ahmad Alnafessah, Suheer Al-Hadhrami, Mai A. Al-Ammar, and Hend S. Al-Khalifa. 2016. Ultra Wideband Indoor Positioning Technologies: Analysis and Recent Advances. Sensors 16, 5 (may 2016), 707. https://doi.org/10.3390/s16050707
- [2] UWB Alliance. 2020. UWB Alliance. Retrieved January 9, 2020 from https://uwballiance.org/
- [3] Kenneth Batstone, Magnus Oskarsson, and Kalle Astrom. 2017. Towards real-time time-of-arrival self-calibration using ultra-wideband anchors. In 2017 International Conference on Indoor Positioning and Indoor Navigation (IPIN). IEEE, Sapporo, 1–8. https://doi.org/10.1109/IPIN.2017.8115885
- [4] Abhimanyu Bhateja, Gurjit Singh Walia, and Rajiv Kapoor. 2016. Non linear state estimation using particle filter based on backtracking search optimization. In 2016 International Conference on Computing, Communication and Automation (ICCCA). IEEE, Greater Noida, India, 342–347. https://doi.org/10.1109/CCAA.2016.7813742
- [5] Saad Biaz and Yiming Ji. 2005. A survey and comparison on localisation algorithms for wireless ad hoc networks. International Journal of Mobile Communications 3, 4 (2005), 374–410.
- [6] Klemen Bregar and Mihael Mohorcic. 2018. Improving Indoor Localization Using Convolutional Neural Networks on Computationally Restricted Devices. IEEE Access 6 (2018), 17429–17441. https://doi.org/10.1109/ACCESS.2018.2817800
- [7] Klemen Bregar, Roman Novak, and Mihael Mohorčič. 2019. Evaluation of range-based indoor tracking algorithms by merging simulation and measurements. EURASIP Journal on Wireless Communications and Networking 2019, 1 (2019), 173.
- [8] R Michael Buehrer, Henk Wymeersch, and Reza Monir Vaghefi. 2018. Collaborative sensor network localization: Algorithms and practical issues. Proc. IEEE 106, 6 (2018), 1089–1114.
- [9] Haofan Cai, Guang Wu, Yifan Chen, and Linshan Jiang. 2015. Indoor collaborative localization method based on ultra-wideband ranging. In 2015 9th European Conference on Antennas and Propagation (EuCAP). IEEE, 1–2.

- [10] Srdjan Čapkun, Maher Hamdi, and Jean-Pierre Hubaux. 2002. GPS-free positioning in mobile ad hoc networks. Cluster Computing 5, 2 (2002), 157–167.
- [11] Cemin Zhang, M. Kuhn, B. Merkl, A.E. Fathy, and M. Mahfouz. 2006. Accurate UWB indoor localization system utilizing time difference of arrival approach. In 2006 IEEE Radio and Wireless Symposium. IEEE, San Diego, CA, USA, 515–518. https://doi.org/10.1109/RWS.2006.1615207
- [12] Leila Chelouah, Fouzi Semchedine, and Louiza Bouallouche-Medjkoune. 2018. Localization protocols for mobile wireless sensor networks: A survey. Computers and Electrical Engineering 71 (oct 2018), 733–751. https://doi.org/10.1016/j.compeleceng.2017.03.024
- [13] Tashnim J.S. Chowdhury, Colin Elkin, Vijay Devabhaktuni, Danda B. Rawat, and Jared Oluoch. 2016. Advances on localization techniques for wireless sensor networks: A survey. , 284–305 pages. https://doi.org/10.1016/j.comnet.2016.10.006
- [14] Angelo Coluccia and Alessio Fascista. 2019. A review of advanced localization techniques for crowdsensing wireless sensor networks. , 988 pages. https://doi.org/10.3390/s19050988
- [15] FiRa Consortium. 2020. FiRa Consortium. Retrieved January 9, 2020 from https://www.firaconsortium.org/
- [16] Jose A Costa, Neal Patwari, and Alfred O Hero III. 2006. Distributed weighted-multidimensional scaling for node localization in sensor networks. ACM Transactions on Sensor Networks (TOSN) 2, 1 (2006), 39–64.
- [17] Marco Crocco, Alessio Del Bue, and Vittorio Murino. 2012. A Bilinear Approach to the Position Self-Calibration of Multiple Sensors. IEEE Transactions on Signal Processing 60, 2 (Feb. 2012), 660–673. https://doi.org/10.1109/TSP.2011.2175387
- [18] Paolo Dabove, Vincenzo Di Pietra, Marco Piras, Ansar Abdul Jabbar, and Syed Ali Kazim. 2018. Indoor positioning using Ultra-wide band (UWB) technologies: Positioning accuracies and sensors' performances. In 2018 IEEE/ION Position, Location and Navigation Symposium (PLANS). IEEE, Monterey, CA, 175–184. https://doi.org/10.1109/PLANS.2018.8373379
- [19] Davide Dardari, Pau Closas, and Petar M. Djuric. 2015. Indoor tracking: Theory, methods, and technologies. IEEE Transactions on Vehicular Technology 64, 4 (apr 2015), 1263–1278. https://doi.org/10.1109/TVT.2015.2403868
- [20] Davide Dardari, Andrea Conti, Jaime Lien, and Moe Z. Win. 2008. The Effect of Cooperation on UWB-Based Positioning Systems Using Experimental Data. EURASIP Journal on Advances in Signal Processing 2008, 1 (Dec. 2008), 513873. https://doi.org/10.1155/2008/513873
- [21] Pavel Davidson and Robert Piché. 2017. A Survey of Selected Indoor Positioning Methods for Smartphones. , 1347–1370 pages. https: //doi.org/10.1109/COMST.2016.2637663
- [22] Brecht De Beelde, Nico Podevijn, David Plets, Emmeric Tanghe, Luc Martens, and Wout Joseph. 2019. Joint Antenna and Channel Modelling for Communication in Metallic Kitchen Environments. In 2019 13th European Conference on Antennas and Propagation (EuCAP). IEEE, 1–4.
- Fabian de Ponte Müller. 2017. Survey on Ranging Sensors and Cooperative Techniques for Relative Positioning of Vehicles. Sensors 17, 2 (Jan. 2017), 271. https://doi.org/10.3390/s17020271
- [24] Decawave Ltd. 2018. DW1000 Metrics for Estimation of Non Line Of Sight Operating Conditions. Technical Report. https://www.decawave.com/wpcontent/uploads/2018/10/APS006\_Part-3-DW1000-Diagnostics-for-NLOS-Channels\_v1.1.pdf Accessed: 2019-11-08.
- [25] Decawave Ltd. 2018. Maximising range in DW1000 based systems. Technical Report. https://www.decawave.com/wp-content/uploads/2018/10/ APS017\_Max-Range-in-DW1000-Systems\_v1.1.pdf Accessed: 2020-01-08.
- [26] Benoit Denis and Norbert Daniele. 2004. NLOS ranging error mitigation in a distributed positioning algorithm for indoor UWB ad-hoc networks. In International Workshop on Wireless Ad-Hoc Networks, 2004. IEEE, 356–360.
- [27] Maria-Gabriella Di Benedetto (Ed.). 2006. UWB communication systems: a comprehensive overview. Number 5 in EURASIP book series on signal processing and communications. Hindawi Publ, New York, NY. OCLC: 254187220.
- [28] Lance Doherty, Laurent El Ghaoui, et al. 2001. Convex position estimation in wireless sensor networks. In Proceedings IEEE INFOCOM 2001. Conference on Computer Communications. Twentieth Annual Joint Conference of the IEEE Computer and Communications Society (Cat. No. 01CH37213), Vol. 3. IEEE, 1655–1663.
- [29] Melike Erol-Kantarci, Hussein T. Mouftah, and Sema Oktug. 2011. A survey of architectures and localization techniques for underwater acoustic sensor networks. , 487–502 pages. https://doi.org/10.1109/SURV.2011.020211.00035
- [30] ETSI. 2019. Ultra Wide Band (UWB) technology. Retrieved September 11, 2019 from https://www.etsi.org/technologies/radio/ultra-wide-band
- [31] European Telecommunications Standards Institute. 2016. Short Range Devices (SRD) using Ultra Wide Band technology (UWB); Harmonised Standard covering the essential requirements of article 3.2 of the Directive 2014/53/EU; Part 2: Requirements for UWB location tracking. Standard. European Telecommunications Standards Institute.
- [32] Andre Filipe Goncalves Ferreira, Duarte Manuel Azevedo Fernandes, Andre Paulo Catarino, and Joao L. Monteiro. 2017. Localization and Positioning Systems for Emergency Responders: A Survey., 2836–2870 pages. https://doi.org/10.1109/COMST.2017.2703620
- [33] Dieter Fox, Wolfram Burgard, Hannes Kruppa, and Sebastian Thrun. 2000. A probabilistic approach to collaborative multi-robot localization. Autonomous robots 8, 3 (2000), 325–344.
- [34] Mohammad Reza Gholami, Luba Tetruashvili, Erik G. Strom, and Yair Censor. 2013. Cooperative Wireless Sensor Network Positioning via Implicit Convex Feasibility. IEEE Transactions on Signal Processing 61, 23 (dec 2013), 5830–5840. https://doi.org/10.1109/TSP.2013.2279770
- [35] Andrea Giorgetti and Marco Chiani. 2013. Time-of-arrival estimation based on information theoretic criteria. IEEE Transactions on Signal Processing 61, 8 (2013), 1869–1879.
- [36] Bernhard Grobwindhager, Carlo Alberto Boano, Michael Rath, and Kay Romer. 2018. Enabling Runtime Adaptation of Physical Layer Settings for Dependable UWB Communications. In 2018 IEEE 19th International Symposium on "A World of Wireless, Mobile and Multimedia Networks" (WoWMoM). IEEE, Chania, Greece, 01–11. https://doi.org/10.1109/WoWMoM.2018.8449776

- [37] Ismail Guvenc and Chia-Chin Chong. 2009. A survey on TOA based wireless localization and NLOS mitigation techniques. IEEE Communications Surveys & Tutorials 11, 3 (2009), 107–124.
- [38] Michael Hamer and Raffaello D'Andrea. 2018. Self-Calibrating Ultra-Wideband Network Supporting Multi-Robot Localization. IEEE Access 6 (2018), 22292–22304. https://doi.org/10.1109/ACCESS.2018.2829020
- [39] Guangjie Han, Huihui Xu, Trung Q. Duong, Jinfang Jiang, and Takahiro Hara. 2013. Localization algorithms of Wireless Sensor Networks: A survey. Telecommunication Systems 52, 4 (apr 2013), 2419–2436. https://doi.org/10.1007/s11235-011-9564-7
- [40] Suining He and S. H.Gary Chan. 2016. Wi-Fi fingerprint-based indoor positioning: Recent advances and comparisons. , 466–490 pages. https://doi.org/10.1109/COMST.2015.2464084
- [41] A. K.M.Mahtab Hossain and Wee Seng Soh. 2015. A survey of calibration-free indoor positioning systems. Computer Communications 66 (jul 2015), 1–13. https://doi.org/10.1016/j.comcom.2015.03.001
- [42] Yi-Lin Hsu, Yan-Ju Chen, and Sheng-Wen Shih. 2016. A Particle Filter Approach for Pedestrian Dead Reckoning Using Wearable Sensors. In 2016 10th International Conference on Innovative Mobile and Internet Services in Ubiquitous Computing (IMIS). IEEE, Fukuoka, Japan, 26–32. https://doi.org/10.1109/IMIS.2016.97
- [43] Alexander T Ihler, John W Fisher, Randolph L Moses, and Alan S Willsky. 2005. Nonparametric belief propagation for self-localization of sensor networks. IEEE Journal on Selected Areas in Communications 23, 4 (2005), 809–819.
- [44] Apple Inc. 2020. Apple iPhone 11. Retrieved February 9, 2020 from https://www.apple.com/iphone-11/
- [45] Beakcheol Jang and Hyunjung Kim. 2019. Indoor positioning technologies without offline fingerprinting map: A survey. IEEE Communications Surveys and Tutorials 21, 1 (jan 2019), 508–525. https://doi.org/10.1109/COMST.2018.2867935
- [46] Tao Jia and R. Michael Buehrer. 2010. On the optimal performance of collaborative position location. IEEE Transactions on Wireless Communications 9, 1 (jan 2010), 374–383. https://doi.org/10.1109/TWC.2010.01.090869
- [47] Young Joon Kim, Byeongho Lee, Hyoungmin So, Dong Hwan Hwang, and Seong Cheol Kim. 2018. Cooperative localization considering estimated location uncertainty in distributed ad hoc networks. *International Journal of Distributed Sensor Networks* 14, 2 (2018), 2018. https: //doi.org/10.1177/1550147718759634
- [48] Hakan Koyuncu and Shuang Hua Yang. 2010. A Survey of Indoor Positioning and Object Locating Systems. Technical Report 5. Loughborough University, UK. 121 pages.
- [49] Michael Kuhn, Cemin Zhang, Brandon Merkl, Depeng Yang, Yazhou Wang, Mohamed Mahfouz, and Aly Fathy. 2008. High accuracy UWB localization in dense indoor environments. In 2008 IEEE International Conference on Ultra-Wideband, Vol. 2. IEEE, 129–132.
- [50] Josef Kulmer, Stefan Hinteregger, Bernhard Grosswindhager, Michael Rath, Mustafa S. Bakr, Erik Leitinger, and Klaus Witrisal. 2017. Using DecaWave UWB transceivers for high-accuracy multipath-assisted indoor positioning. In 2017 IEEE International Conference on Communications Workshops (ICC Workshops). IEEE, Paris, France, 1239–1245. https://doi.org/10.1109/ICCW.2017.7962828
- [51] Christos Laoudias, Adriano Moreira, Sunwoo Kim, Sangwoo Lee, Lauri Wirola, and Carlo Fischione. 2018. A survey of enabling technologies for network localization, tracking, and navigation. , 3607–3644 pages. https://doi.org/10.1109/COMST.2018.2855063
- [52] E.G. Larsson. 2004. CramÉr-Rao Bound Analysis of Distributed Positioning in Sensor Networks. IEEE Signal Processing Letters 11, 3 (mar 2004), 334–337. https://doi.org/10.1109/LSP.2003.822899
- [53] Yong Up Lee. 2011. Weighted-average based aoa parameter estimations for LR-UWB wireless positioning system. IEICE transactions on communications 94, 12 (2011), 3599–3602.
- [54] Bin Li, Nan Wu, Hua Wang, Desheng Shi, Weijie Yuan, and Jingming Kuang. 2013. Particle swarm optimization-based particle filter for cooperative localization in wireless networks. In 2013 International Conference on Wireless Communications and Signal Processing. IEEE, 1–6.
- [55] Shenghong Li, Mark Hedley, and Iain B. Collings. 2015. New Efficient Indoor Cooperative Localization Algorithm With Empirical Ranging Error Model. IEEE Journal on Selected Areas in Communications 33, 7 (jul 2015), 1407–1417. https://doi.org/10.1109/JSAC.2015.2430273
- [56] Shenghong Li, Wei Ni, Chang Kyung Sung, and Mark Hedley. 2018. Recent advances on cooperative wireless localization and their application in inhomogeneous propagation environments. *Computer Networks* 142 (Sept. 2018), 253–271. https://doi.org/10.1016/j.comnet.2018.06.017
- [57] Rejina Wei Choi Ling, Ankur Gupta, Ankush Vashistha, Manmohan Sharma, and Choi Look Law. 2018. High precision UWB-IR indoor positioning system for IoT applications. In 2018 IEEE 4th World Forum on Internet of Things (WF-IoT). IEEE, Singapore, 135–139. https://doi.org/10.1109/WF-IoT.2018.8355162
- [58] Hui Liu, Houshang Darabi, Pat Banerjee, and Jing Liu. 2007. Survey of Wireless Indoor Positioning Techniques and Systems. IEEE Transactions on Systems, Man and Cybernetics, Part C (Applications and Reviews) 37, 6 (Nov. 2007), 1067–1080. https://doi.org/10.1109/TSMCC.2007.905750
- [59] Nicola Macoir, Jan Bauwens, Bart Jooris, Ben Van Herbruggen, Jen Rossey, Jeroen Hoebeke, and Eli De Poorter. 2019. Uwb localization with battery-powered wireless backbone for drone-based inventory management. Sensors 19, 3 (2019), 467.
- [60] Nicola Macoir, Matteo Ridolfi, Jen Rossey, Ingrid Moerman, and Eli De Poorter. 2018. MAC Protocol for Supporting Multiple Roaming Users in Mult-Cell UWB Localization Networks. In 2018 IEEE 19th International Symposium on "A World of Wireless, Mobile and Multimedia Networks" (WoWMoM). IEEE, Chania, Greece, 588–599. https://doi.org/10.1109/WoWMoM.2018.8449815
- [61] Halgurd S. Maghdid, Ihsan Alshahib Lami, Kayhan Zrar Ghafoor, and Jaime Lloret. 2016. Seamless Outdoors-Indoors Localization Solutions on Smartphones. Comput. Surveys 48, 4 (feb 2016), 1–34. https://doi.org/10.1145/2871166
- [62] Mohamed R Mahfouz, Cemin Zhang, Brandon C Merkl, Michael J Kuhn, and Aly E Fathy. 2008. Investigation of high-accuracy indoor 3-D positioning using UWB technology. *IEEE Transactions on Microwave Theory and Techniques* 56, 6 (2008), 1316–1330.

- [63] Paul Meissner, Thomas Gigl, and Klaus Witrisal. 2010. UWB sequential Monte Carlo positioning using virtual anchors. In 2010 International Conference on Indoor Positioning and Indoor Navigation. IEEE, 1–10. https://doi.org/10.1109/IPIN.2010.5647797
- [64] Zemene Walle Mekonnen and Armin Wittneben. 2015. Self-calibration method for TOA based localization systems with generic synchronization requirement. In 2015 IEEE International Conference on Communications (ICC). IEEE, London, 4618–4623. https://doi.org/10.1109/ICC.2015.7249051
- [65] Hetal P. Mistry and Nital H. Mistry. 2015. RSSI Based Localization Scheme in Wireless Sensor Networks: A Survey. In 2015 Fifth International Conference on Advanced Computing & Communication Technologies. IEEE, Haryana, India, 647–652. https://doi.org/10.1109/ACCT.2015.105
- [66] Hessam Mohammadmoradi and Omprakash Gnawali. 2019. Study and Mitigation of Non-Cooperative UWB Interference on Ranging. In Proceedings of the 2019 International Conference on Embedded Wireless Systems and Networks (Beijing, China) (EWSN '19). Junction Publishing, USA, 142-153.
- [67] Hessam Mohammadmoradi, Milad Heydariaan, and Omprakash Gnawali. 2018. UWB Physical Layer Adaptation for Best Ranging Performance within Application Constraints. In Proceedings of the 2nd International Conference on Smart Digital Environment. ACM, 119–126.
- [68] Esmond Mok, Linyuan Xia, Guenther Retscher, and Hui Tian. 2010. A case study on the feasibility and performance of an UWB-AoA real time location system for resources management of civil construction projects. *Journal of Applied Geodesy* 4, 1 (2010), 23–32.
- [69] Randolph L Moses, Dushyanth Krishnamurthy, and Robert M Patterson. 2003. A self-localization method for wireless sensor networks. EURASIP Journal on Advances in Signal Processing 2003, 4 (2003), 839843.
- [70] Ville Niemela, Jussi Haapola, Matti Hamalainen, and Jari Iinatti. 2017. An Ultra Wideband Survey: Global Regulations and Impulse Radio Research Based on Standards. IEEE Communications Surveys & Tutorials 19, 2 (2017), 874–890. https://doi.org/10.1109/COMST.2016.2634593
- [71] Youngtae Noh, Hirozumi Yamaguchi, and Uichin Lee. 2016. Infrastructure-free collaborative indoor positioning scheme for time-critical team operations. IEEE Transactions on Systems, Man, and Cybernetics: Systems 48, 3 (2016), 418–432.
- [72] Abdelmoumen Norrdine. 2012. An algebraic solution to the multilateration problem. In Proceedings of the 15th international conference on indoor positioning and indoor navigation, Sydney, Australia, Vol. 1315.
- [73] Sreenivasulu Pala, Sarada Jayan, and Dhanesh G Kurup. 2020. An accurate UWB based localization system using modified leading edge detection algorithm. Ad Hoc Networks 97 (2020), 102017.
- [74] Zhibo Pang, Binbin Xie, Hui Zhu, and Michele Luvisotto. 2019. Location aided commissioning of building automation devices enabled by high accuracy indoor positioning. Journal of Industrial Information Integration 15 (2019), 258–263.
- [75] Neal Patwari, Joshua N Ash, Spyros Kyperountas, Alfred O Hero, Randolph L Moses, and Neiyer S Correal. 2005. Locating the nodes: cooperative localization in wireless sensor networks. IEEE Signal processing magazine 22, 4 (2005), 54–69.
- [76] Neal Patwari, Alfred O Hero, Matt Perkins, Neiyer S Correal, and Robert J O'dea. 2003. Relative location estimation in wireless sensor networks. IEEE Transactions on signal processing 51, 8 (2003), 2137–2148.
- [77] Anup Paul and Takuro Sato. 2017. Localization in wireless sensor networks: a survey on algorithms, measurement techniques, applications and challenges. Journal of Sensor and Actuator Networks 6, 4 (2017), 24.
- [78] Andreas De Preter, Glenn Goysens, Jan Anthonis, Jan Swevers, and Goele Pipeleers. 2019. Range Bias Modeling and Autocalibration of an UWB Positioning System. In 2019 International Conference on Indoor Positioning and Indoor Navigation (IPIN). IEEE, Pisa, Italy, 1–8. https: //doi.org/10.1109/IPIN.2019.8911815
- [79] Nissanka B Priyantha, Hari Balakrishnan, Erik Demaine, and Seth Teller. 2003. Anchor-free distributed localization in sensor networks. (2003), 340–341.
- [80] AK Rajendra, Y Song, K Gururaj, CX Wang, GF Cai, CL Law, WP Tay, and HP Chow. 2017. UWB based real-time cooperative localization system. In 2017 International Conference on Indoor Positioning and Indoor Navigation (IPIN).
- [81] Matteo Ridolfi, Nicola Macoir, Jono Vanhie-Van Gerwen, Jen Rossey, Jeroen Hoebeke, and Eli de Poorter. 2019. Testbed for warehouse automation experiments using mobile AGVs and drones. In IEEE INFOCOM 2019 - IEEE Conference on Computer Communications Workshops (INFOCOM WKSHPS). IEEE, Paris, France, 919–920. https://doi.org/10.1109/INFCOMW.2019.8845218
- [82] Matteo Ridolfi, Samuel Van de Velde, Heidi Steendam, and Eli De Poorter. 2016. WiFi ad-hoc mesh network and MAC protocol solution for UWB indoor localization systems. In 2016 Symposium on Communications and Vehicular Technologies (SCVT). IEEE, Mons, Belgium, 1–6. https://doi.org/10.1109/SCVT.2016.7797661
- [83] Matteo Ridolfi, Samuel Van de Velde, Heidi Steendam, and Eli De Poorter. 2018. Analysis of the scalability of UWB indoor localization solutions for high user densities. Sensors 18, 6 (2018), 1875.
- [84] Matteo Ridolfi, Stef Vandermeeren, Jense Defraye, Heidi Steendam, Joeri Gerlo, Dirk De Clercq, Jeroen Hoebeke, and Eli De Poorter. 2018. Experimental evaluation of UWB indoor positioning for sport postures. Sensors 18, 1 (2018), 168.
- [85] Thuraiappah Sathyan and Mark Hedley. 2012. Fast and accurate cooperative tracking in wireless networks. IEEE Transactions on Mobile Computing 12, 9 (2012), 1801–1813.
- [86] Andreas Savvides, Chih-Chieh Han, and Mani B. Strivastava. 2001. Dynamic fine-grained localization in Ad-Hoc networks of sensors. In Proceedings of the 7th annual international conference on Mobile computing and networking - MobiCom '01. ACM Press, Rome, Italy, 166–179. https://doi.org/10.1145/381677.381693
- [87] Petr Sedlacek, Martin Slanina, and Pavel Masek. 2019. An Overview of the IEEE 802.15.4z Standard its Comparison and to the Existing UWB Standards. In 2019 29th International Conference Radioelektronika (RADIOELEKTRONIKA). IEEE, Pardubice, Czech Republic, 1–6. https://doi.org/10. 1109/RADIOELEK.2019.8733537

- [88] Guowei Shen, Rudolf Zetik, Honghui Yan, Ole Hirsch, and Reiner S Thomä. 2010. Time of arrival estimation for range-based localization in UWB sensor networks. In 2010 IEEE International Conference on Ultra-Wideband, Vol. 2. IEEE, 1–4.
- [89] Guowei Shi and Ying Ming. 2016. Survey of Indoor Positioning Systems Based on Ultra-wideband (UWB) Technology. In Wireless Communications, Networking and Applications, Qing-An Zeng (Ed.). Springer India, New Delhi, 1269–1278.
- [90] Qin Shi, Sihao Zhao, Xiaowei Oui, Mingquan Lu, and Mengdi Jia. 2019. Anchor self-localization algorithm based on UWB ranging and inertial measurements. *Tsinghua Science and Technology* 24, 6 (2019), 728–737.
- [91] Rathin Chandra Shit, Suraj Sharma, Deepak Puthal, and Albert Y. Zomaya. 2018. Location of Things (LoT): A review and taxonomy of sensors localization in IoT infrastructure. *IEEE Communications Surveys and Tutorials* 20, 3 (jul 2018), 2028–2061. https://doi.org/10.1109/COMST.2018. 2798591
- [92] Juri Sidorenko, Volker Schatz, Norbert Scherer-Negenborn, Michael Arens, and Urs Hugentobler. 2019. Decawave UWB Clock Drift Correction and Power Self-Calibration. Sensors 19, 13 (July 2019), 2942. https://doi.org/10.3390/s19132942
- [93] Ritesh Kumar Singh, Michiel Aernouts, Mats De Meyer, Maarten Weyn, and Rafael Berkvens. 2020. Leveraging LoRaWAN Technology for Precision Agriculture in Greenhouses. Sensors 20, 7 (2020), 1827.
- [94] Santar Pal Singh and SC Sharma. 2015. A survey on cluster based routing protocols in wireless sensor networks. Procedia computer science 45 (2015), 687–695.
- [95] Ananth Subramanian. 2007. UWB linear quadratic frequency domain frequency invariant beamforming and angle of arrival estimation. In 2007 IEEE 65th Vehicular Technology Conference-VTC2007-Spring. IEEE, 614–618.
- [96] Slavisa Tomic, Marko Beko, Rui Dinis, and Miroslava Raspopovic. 2014. Distributed RSS-based localization in wireless sensor networks using convex relaxation. In 2014 International Conference on Computing, Networking and Communications (ICNC). IEEE, 853–857.
- [97] Jens Trogh, David Plets, Luc Martens, and Wout Joseph. 2015. Advanced real-time indoor tracking based on the viterbi algorithm and semantic data. International Journal of Distributed Sensor Networks 11, 10 (2015), 271818.
- [98] T.K.K. Tsang and M.N. El-Gamal. 2005. Ultra-wideband (UWB) communications systems: an overview. In *The 3rd International IEEE-NEWCAS Conference*, 2005. IEEE, Quebec City, Canada, 124–129. https://doi.org/10.1109/NEWCAS.2005.1496688
- [99] Reza Monir Vaghefi and R. Michael Buehrer. 2017. Cooperative Source Node Tracking in Non-Line-of-Sight Environments. IEEE Transactions on Mobile Computing 16, 5 (may 2017), 1287–1299. https://doi.org/10.1109/TMC.2016.2591540
- [100] Samuel Van de Velde and Heidi Steendam. 2012. CUPID algorithm for cooperative indoor multipath-aided localization. In 2012 International Conference on Indoor Positioning and Indoor Navigation (IPIN). IEEE, 1–6. https://doi.org/10.1109/IPIN.2012.6418912
- [101] Samuel Van de Velde, Henk Wymeersch, and Heidi Steendam. 2012. Comparison of message passing algorithms for cooperative localization under NLOS conditions. In 2012 9th Workshop on Positioning, Navigation and Communication. IEEE, 1–6.
- [102] Tom Van Haute, Eli De Poorter, Filip Lemic, Vlado Handziski, Niklas Wirström, Thiemo Voigt, Adam Wolisz, and Ingrid Moerman. 2015. Platform for benchmarking of RF-based indoor localization solutions. *IEEE Communications Magazine* 53, 9 (2015), 126–133.
- [103] Ankush Vashistha, Ankur Gupta, and Choi Look Law. 2018. Self calibration of the anchor nodes for UWB-IR TDOA based indoor positioning system. In 2018 IEEE 4th World Forum on Internet of Things (WF-IoT). IEEE, Singapore, 688–693. https://doi.org/10.1109/WF-IoT.2018.8355163
- [104] Davide Vecchia, Pablo Corbalan, Timofei Istomin, and Gian Pietro Picco. 2019. TALLA: Large-scale TDoA Localization with Ultra-wideband Radios. In 2019 International Conference on Indoor Positioning and Indoor Navigation (IPIN). IEEE, Pisa, Italy, 1–8. https://doi.org/10.1109/IPIN.2019.8911790
- [105] Moe Z Win, Andrea Conti, Santiago Mazuelas, Yuan Shen, Wesley M Gifford, Davide Dardari, and Marco Chiani. 2011. Network localization and navigation via cooperation. *IEEE Communications Magazine* 49, 5 (2011), 56–62.
- [106] Woo Cheol Chung and Dong Sam Ha. 2003. An accurate ultra wideband (uwb) ranging for precision asset location. In IEEE Conference on Ultra Wideband Systems and Technologies, 2003. IEEE, Reston, Virginia, USA, 389–393. https://doi.org/10.1109/UWBST.2003.1267870
- [107] Henk Wymeersch, Jaime Lien, and Moe Z. Win. 2009. Cooperative Localization in Wireless Networks. Proc. IEEE 97, 2 (feb 2009), 427–450. https://doi.org/10.1109/JPROC.2008.2008853
- [108] Jun Xu, Maode Ma, and Choi Look Law. 2008. AOA cooperative position localization. In IEEE GLOBECOM 2008-2008 IEEE Global Telecommunications Conference. IEEE, 1–5.
- [109] Liyuan Xu, Lin Yao, Jie He, Peng Wang, Keping Long, and Qin Wang. 2018. Collaborative Geolocation Based on Imprecise Initial Coordinates for Internet of Things. IEEE Access 6 (2018), 48850–48858.
- [110] Nikolay Yakovenko, Riad Ghabra, and Chadi Shaya. 2019. System and methods for vehicle passive keyless entry triggered by smartphone proximity detection. US Patent 10,328,900.
- [111] Ali Yassin, Youssef Nasser, Mariette Awad, Ahmed Al-Dubai, Ran Liu, Chau Yuen, Ronald Raulefs, and Elias Aboutanios. 2017. Recent Advances in Indoor Localization: A Survey on Theoretical Approaches and Applications. *IEEE Communications Surveys & Tutorials* 19, 2 (2017), 1327–1346. https://doi.org/10.1109/COMST.2016.2632427
- [112] Feng Yin, Carsten Fritsche, Di Jin, Fredrik Gustafsson, and Abdelhak M Zoubir. 2015. Cooperative localization in WSNs using Gaussian mixture modeling: Distributed ECM algorithms. *IEEE Transactions on Signal Processing* 63, 6 (2015), 1448–1463.
- [113] Wentao Yu, Xin Zhao, and Guangyi Sun. 2017. Self-calibration of anchor positions for indoor localization. In 2017 IEEE International Conference on Robotics and Biomimetics (ROBIO). IEEE, Macau, 581–586. https://doi.org/10.1109/ROBIO.2017.8324479
- [114] Faheem Zafari, Athanasios Gkelias, and Kin K. Leung. 2019. A Survey of Indoor Localization Systems and Technologies. IEEE Communications Surveys & Tutorials 21, 3 (2019), 2568–2599. https://doi.org/10.1109/COMST.2019.2911558

- [115] Giovanni Zanca, Francesco Zorzi, Andrea Zanella, and Michele Zorzi. 2008. Experimental comparison of RSSI-based localization algorithms for indoor wireless sensor networks. In Proceedings of the workshop on Real-world wireless sensor networks - REALWSN '08. ACM Press, Glasgow, Scotland, 1. https://doi.org/10.1145/1435473.1435475
- [116] Da Zhang, Feng Xia, Zhuo Yang, Lin Yao, and Wenhong Zhao. 2010. Localization technologies for indoor human tracking. In 2010 5th International Conference on Future Information Technology, FutureTech 2010 - Proceedings. https://doi.org/10.1109/FUTURETECH.2010.5482731
- [117] Jinyun Zhang, Philip V Orlik, Zafer Sahinoglu, Andreas F Molisch, and Patrick Kinney. 2009. UWB systems for wireless sensor networks. Proc. IEEE 97, 2 (2009), 313–331.
- [118] Bingpeng Zhou and Qingchun Chen. 2016. On the Particle-assisted Stochastic Search Mechanism in Wireless Cooperative Localization. IEEE Transactions on Wireless Communications (2016), 1–1. https://doi.org/10.1109/TWC.2016.2545665
- [119] Chunsheng Zhu, Victor CM Leung, Laurence T Yang, and Lei Shu. 2014. Collaborative location-based sleep scheduling for wireless sensor networks integrated with mobile cloud computing. *IEEE Trans. Comput.* 64, 7 (2014), 1844–1856.
- [120] Xiuyan Zhu, Yuan Feng, et al. 2013. RSSI-based algorithm for indoor localization. Communications and Network 5, 02 (2013), 37.
- [121] Xiuming Zhu, Pei-Chi Huang, Song Han, Aloysius K Mok, Deji Chen, and Mark Nixon. 2012. RoamingHART: A collaborative localization system on WirelessHART. In 2012 IEEE 18th Real Time and Embedded Technology and Applications Symposium. IEEE, 241–250.

28